Author’s response to reviews

Title: Differences in primary health care use among sub-Saharan African immigrants in Norway: a register-based study

Authors:

Esperanza Diaz (esperanza.diaz@uib.no)

Vivian Mbanya (vivimbanya@yahoo.com)

Abdi A Gele (abdi@nakmi.no)

Bernadette Kumar (bernadette.kumar@nakmi.no)

Version: 2 Date: 20 Jun 2017

Author’s response to reviews:

Dear Editor,

The manuscript has been revised as requested by the editors.

Regards,

Mbanya Vivian